Author's response to reviews

Title: Characterization of medical students recall of factual knowledge using learning objects and repeated testing in a novel e-learning system

Authors:

Tiago Taveira-Gomes (tiago.taveira@me.com)
Rui Prado-Costa (rcosta4540@gmail.com)
Milton Severo (milton@med.up.pt)
Maria Amélia Ferreira (mameliaferreira@med.up.pt)

Version: 4 Date: 22 November 2014

Author's response to reviews:

Dear Editors,

We are very grateful for having accepted our work for publication.

In accordance to required changes, we have shortened the abstract of our manuscript to less than 350 words, and included it into the manuscript details abstract section.

No changes were made to the files submitted on the previous revision, as requested.

Please let me know if further revisions are required,

Best regards,

Tiago Taveira-Gomes